PLOS MEDICINE



Citation: Norris T, Cole TJ, Bann D, Harner M, Hardy R, Li L, et al. (2020) Duration of obesity exposure between ages 10 and 40 years and its relationship with cardiometabolic disease risk factors: A cohort study. PLoS Med 17(12): e1003387. https://doi.org/10.1371/journal. pmed.1003387

Academic Editor: Jason Wu, George Institute for Global Health, AUSTRALIA

Received: March 19, 2020

Accepted: October 1, 2020

Published: December 8, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003387

Copyright: © 2020 Norris et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The dataset used in this research uses harmonised data from three cohort studies. The original and the harmonised

RESEARCH ARTICLE

Duration of obesity exposure between ages 10 and 40 years and its relationship with cardiometabolic disease risk factors: A cohort study

Tom Norris^{1*}, Tim J. Cole², David Bann³, Mark Hamer⁴, Rebecca Hardy⁵, Leah Li², Ken K. Ong⁶, George B. Ploubidis³, Russell Viner², William Johnson¹

School of Sport Exercise and Health Sciences, Loughborough University, Loughborough, United Kingdom,
UCL Great Ormond Street Institute of Child Health, London, United Kingdom,
Centre for Longitudinal Studies, Department of Social Science, University College London, London, United Kingdom,
Division of Surgery & Interventional Science/Institute of Sport, Exercise and Health, UCL, London, United Kingdom,
UCL Institute of Education, London, United Kingdom,
MRC Epidemiology Unit, University of Cambridge, Cambridge, United Kingdom

* t.norris@lboro.ac.uk

Abstract

Background

Individuals with obesity do not represent a homogeneous group in terms of cardiometabolic risk. Using 3 nationally representative British birth cohorts, we investigated whether the duration of obesity was related to heterogeneity in cardiometabolic risk.

Methods and findings

We used harmonised body mass index (BMI) and cardiometabolic disease risk factor data from 20,746 participants (49.1% male and 97.2% white British) enrolled in 3 British birth cohort studies: the 1946 National Survey of Health and Development (NSHD), the 1958 National Child Development Study (NCDS), and the 1970 British Cohort Study (BCS70). Within each cohort, individual life course BMI trajectories were created between 10 and 40 years of age, and from these, age of obesity onset, duration spent obese (range 0 to 30 years), and cumulative obesity severity were derived. Obesity duration was examined in relation to a number of cardiometabolic disease risk factors collected in mid-adulthood: systolic (SBP) and diastolic blood pressure (DBP), high-density-lipoprotein cholesterol (HDL-C), and glycated haemoglobin (HbA1c).

A greater obesity duration was associated with worse values for all cardiometabolic disease risk factors. The strongest association with obesity duration was for HbA1c: HbA1c levels in those with obesity for <5 years were relatively higher by 5% (95% CI: 4, 6), compared with never obese, increasing to 20% (95% CI: 17, 23) higher in those with obesity for 20 to 30 years. When adjustment was made for obesity severity, the association with obesity duration was largely attenuated for SBP, DBP, and HDL-C. For HbA1c, however, the association with obesity duration persisted, independent of obesity severity. Due to pooling of 3 1946 NSHD data are made available to researchers who submit data requests to mrclha.swiftinfo@ucl. ac.uk; see also the full policy documents at http:// www.nshd.mrc.ac.uk/data.aspx. The original data for the 1958 NCDS and 1970 BCS are available from the UK Data Service (https://ukdataservice.ac. uk/); applications for access to any data (original or harmonized) held by the UK Data Archive that forms part of the NCDS Biomedical Resource will require special license and should be submitted to clsfeedback@ioe.ac.uk.

Funding: WJ is supported by a UK Medical Research Council (MRC) New Investigator Research Grant (MR/P023347/1) (https://mrc.ukri. org/). WJ acknowledges support from the National Institute for Health Research (NIHR) Leicester Biomedical Research Centre, which is a partnership between University Hospitals of Leicester NHS Trust, Loughborough University, and the University of Leicester. DB is supported by the Economic and Social Research Council (grant number ES/ M001660/1) (https://esrc.ukri.org/) and The Academy of Medical Sciences / Wellcome Trust ("Springboard Health of the Public in 2040" award: HOP001/1025) (https://wellcome.ac.uk/). RH is Director of the CLOSER consortium which is funded by the Economic and Social Research Council (award reference: ES/K000357/1) (https:// esrc.ukri.org/). TJC was supported by MRC research grant MR/R010692/1 (https://mrc.ukri. org/). GBP is supported by the Economic and Social Research Council ES/M001660/1 (https:// esrc.ukri.org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: AUC, area under the curve; BCS70, 1970 British Cohort Study; BMI, body mass index; BIC, Bayesian information criterion; BP, blood pressure; DBP, diastolic blood pressure; FMI, fraction missing information; HbA1c, glycated haemoglobin; HDL-C, high-density-lipoprotein cholesterol; IOTF, International Obesity Task Force; IQR, interquartile range; MICE, multiple imputation by chained equations; NCDS, National Child Development Study; NSHD, National Survey of Health and Development; RR, relative risk; SBP, systolic blood pressure; SEP, socioeconomic position; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. cohorts and thus the availability of only a limited number harmonised variables across cohorts, our models included adjustment for only a small number of potential confounding variables, meaning there is a possibility of residual confounding.

Conclusions

Given that the obesity epidemic is characterised by a much earlier onset of obesity and consequently a greater lifetime exposure, our findings suggest that health policy recommendations aimed at preventing early obesity onset, and therefore reducing lifetime exposure, may help reduce the risk of diabetes, independently of obesity severity. However, to test the robustness of our observed associations, triangulation of evidence from different epidemiological approaches (e.g., mendelian randomization and negative control studies) should be obtained.

Author summary

Why was this study done?

- People with obesity (body mass index (BMI) $> 30 \text{ kg/m}^2$) do not all share the same risk for development of cardiometabolic disease risk factors.
- The duration a person has spent with obesity over their life course could be 1 factor contributing to the variation observed in cardiometabolic risk.
- However, previous studies have been unable to adequately separate the effects of obesity duration (how long a person has been obese) and obesity severity (the magnitude of a person's BMI).

What did the researchers do and find?

- We derived BMI trajectories between 10 and 40 years of age in 20,746 participants and calculated each person's total time spent with obesity (duration) as well as their severity of obesity.
- We related obesity duration to cardiometabolic disease risk factors (SBP and DBP, HDL-C, and HbA1c) in mid-adulthood.
- A greater obesity duration was associated with worse values for all cardiometabolic disease risk factors. The strongest association with obesity duration was for HbA1c: HbA1c levels in those with obesity for <5 years were relatively higher by 5% (95% CI: 4, 6), compared with never obese, increasing to 20% (95% CI: 17, 23) higher in those with obesity for 20 to 30 years.
- This positive association between obesity duration and cardiometabolic disease risk factors was largely attenuated when adjusting for obesity severity, except for HbA1c.

What do these findings mean?

- The obesity epidemic is characterised by trends towards earlier onset and consequently greater lifetime exposure.
- Our findings are important as they suggest that health policy recommendations aimed at preventing early onset obesity, and therefore reducing lifetime obesity exposure, may help reduce the risk for diabetes.
- However, due to the small number of potential confounding variables we were able to include in our analysis, the contribution of residual confounding to our findings should be acknowledged. Furthermore, the robustness of the observed associations should be tested using different markers of glucose metabolism and triangulated using different epidemiological approaches underpinned by different assumptions and sources of bias (e.g. mendelian randomization and negative control studies).

Introduction

Obesity is a global public health concern. Worldwide prevalence of child and adolescent obesity (defined according to a body mass index [BMI] of >2 standard deviations above age-specific World Health Organization cut-offs) has increased from 0.9% and 0.7% in boys and girls, respectively, in 1975 to 7.8% and 5.6%, respectively, in 2016. These increases in child obesity accompany significant increases in global adult obesity, with prevalence increasing from 3% and 6.6% of males and females, respectively, in 1975 to 11.6% and 15.7%, respectively, in 2016 (defined according to a BMI of $>30 \text{ kg/m}^2$) [1]. While this epidemic is associated with many adverse health outcomes, particularly cardiovascular disease-related morbidity and mortality [2], individuals with obesity do not represent a homogeneous group in terms of cardiometabolic risk. Indeed, there exists a group of individuals who, while exceeding the standard BMI cut-off for obesity (\geq 30 kg/m²), are regarded as metabolically healthy because they have an absence of other major cardiovascular risk factors. The life course traits contributing to this heterogeneity in cardiometabolic risk have received little attention, but it is likely that a large proportion of the heterogeneity is related, in particular, to the length of time a person spends obese [3,4]. It has been demonstrated that younger individuals are now accumulating greater exposure to overweight or obesity throughout their lives [5], so a comprehensive understanding of the influence of the duration of obesity on the development of cardiometabolic risk factors is critical.

Abraham and colleagues [6] published one of the first studies investigating this heterogeneity in cardiometabolic risk for a given weight, observing that rates of some cardiovascular diseases were highest among individuals who were most overweight in adulthood but below the average weight in childhood. As this study, and others which have replicated that analysis [7– 10], are based on weight status at just 2 time points however, obesity duration can be estimated only crudely. More frequent longitudinal measurements of weight are required for a fuller picture. Furthermore, a detailed measurement schedule is also required in order to differentiate between obesity duration, the age of obesity onset, and the severity of obesity, as these, although correlated, may confer different health risks [11,12]. For example, due to the changes in insulin sensitivity that occur during pubertal development [13], an obesity onset in adolescence may be more deleterious for insulin resistance and diabetes than an onset during another period of the life course.

A handful of studies with such data have observed positive associations between obesity duration and several cardiometabolic disease risk factors including metabolic syndrome [11,14], hypertriglyceridemia [14], dyslipidaemia [14,15], and blood pressure [16]. Most evidence relate to the association with type 2 diabetes, however, with numerous studies observing a positive relationship with obesity duration [15,17-23]. The largest of these studies (n = 61, 821) [21] observed that for each 2-year increment in obesity duration, the risk of type 2 diabetes increased by 14%, although, as observed in other studies [19,22], estimates were attenuated upon adjustment for current weight (representing obesity severity). However, these studies have important limitations, including retrospective designs [14,15], categorising the outcome variable (thus ignoring the observed distribution) [14,15,17,20,21], an a priori assumption of a linear relationship between obesity duration and outcomes [17,18], and assuming that once a person becomes obese they remain obese, thus removing the possibility for weight cycling [14,15,21–23]. Another important limitation is the adjustment of the obesity duration-outcome relationship for current BMI (i.e., at outcome assessment) in order to separate the effects of obesity duration and severity [15,18,20-22]. BMI at outcome assessment does not capture the true extent of obesity severity as it ignores (potentially greater) severity occurring at earlier time points. For example, consider 2 adults, adult A and adult B, who both have a BMI of 35 kg/m^2 at follow-up and who have both been obese for 20 years. Adult A has had a constant BMI of 35 kg/m², while adult B has had a BMI as high as 45 kg/m² during this period. It is unlikely that the cardiometabolic health risks associated with these 2 profiles are homogeneous.

Recently, the concept of "obese-years" has been proposed, which combines the degree and duration of obesity into a single variable [24,25]. In the study by Araujo and colleagues [25], an area under the curve (AUC) of BMI (BMI_{AUC}) was used to summarize duration and severity of BMI. It is possible however to obtain a mean obesity severity over any period by dividing this AUC by obesity duration, thus separating the effects of severity and duration. To our knowledge, no study has done this, and thus, robust evidence of the association between obesity duration and cardiometabolic risk factors, which is truly independent of obesity severity, is lacking.

Using data from 3 British birth cohort studies, the aim of the present study was to model serial measurements of BMI obtained across the life course in order to derive, for each individual, the following obesity traits: duration of obesity exposure between ages 10 and 40 years, age of obesity onset, and obesity severity. These parameters were then used to relate obesity duration, with and without adjustment for obesity severity, to systolic (SBP) and diastolic blood pressure (DBP), high-density lipoprotein cholesterol (HDL-C), and glycated haemoglobin (HbA1c) in mid-adulthood.

Methods

Samples

The 3 British birth cohort studies used in these analyses have been previously described in detail elsewhere [26–28] and were designed to be nationally representative when initiated. The MRC National Survey of Health and Development (NSHD) was initiated in 1946 and recruited 5,362 participants. The National Child Development Study (NCDS) was initiated in 1958 and recruited 17,416 participants. The 1970 British Cohort Study (BCS70) was initiated in 1970 and recruited 16,571 participants.

Ethics statement

All of the studies have received ethical approval and obtained informed parental and/or participant consent, both of which cover the secondary analyses reported here. Data collection in the NSHD received multicentre research ethics committee approval (MREC98/1/121), the NCDS obtained ethical approval from the South East MREC (ref:01/1/44) and the BCS70 obtained ethical approval from the National Research Ethics Service (NRES) Committee South East Coast–Brighton and Sussex (Ref. 15/LO/1446). Further details are available from the study websites and/or cohort profiles [26–29].

For this analysis, we identified a target sample of 20,746 (NSHD: n = 2,968; NCDS: n = 9,302; BCS: n = 8,476) participants who attended the biomedical sweep where cardiometabolic disease risk factor data were collected (see below) and contributed BMI data for the derivation of our exposure variable: obesity duration (S1 Fig).

Serial BMI data

As described elsewhere [30], serial BMI (kg/m²) was derived and harmonised in each study from measured or self-reported weight and height collected at the target ages 11, 15, 20 (self-report), 26 (self-report), 36, and 43 years in the 1946 NSHD; 11, 16, 23 (self-report), 33, and 42 (self-report) years in the 1958 NCDS; and 10, 16 (one-third self-report), 26 (self-report), 30 (self-report), 34 (self-report), and 42 (self-report) years in the 1970 BCS.

There were 21,009 observations of BMI from 4,702 participants in the 1946 cohort, with 74% of the sample having 4 or more observations. There were 57,545 observations of BMI from 16.274 participants in the 1958 cohort, with 80% of the sample having 3 or more observations. Finally, there were 56,275 observations of BMI from 15,437 participants in the 1970 cohort, with 72% of the sample having 3 or more observations.

Cardiometabolic disease risk factors in adulthood

In each cohort, a biomedical sweep, with venous blood sampling was conducted in adulthood, at 53 years in the 1946 cohort (n = 3 053), 44 years in the 1958 cohort (n = 9 377), and 46 years in the 1970 cohort (n = 8 581). Measurements of SBP and DBP were obtained as well as blood cardiometabolic biomarkers (HbA1c and HDL-C). More information about the measurement protocols can be found in S1 Text.

Statistical analysis

TN and WJ determined which analyses to perform and include in the present paper in January 2019 after discussing options with all coauthors. The analysis plan was revised in May (modelling obesity duration as a categorical variable rather than a continuous variable) and October 2019 (removing LDL-cholesterol as an outcome due to high amount of missing data) when further exposure and outcome data were obtained and explored. Further analyses were added in June 2020 in response to reviewer comments (adjusting for further putative confounding variables in the regression models, adding *sexXduration* interaction models in the supplementary analyses).

Obesity duration parameters. In order to identify obesity and derive obesity parameters throughout the life course, we modelled individual child-adulthood trajectories of BMI from 10 to 40 years of age. These life course BMI trajectories were modelled within each cohort separately, due to the previously described between-cohort heterogeneity in the age-related progression of obesity from childhood to adulthood [5]. Models included all participants who contributed at least 1 measurement of BMI during the studied age range (NSHD: 11 to 43

years; NCDS: 11 to 42 years; BCS: 10 to 42 years). The BMI trajectories were modelled using restricted cubic splines with mixed effects, with measurement occasion at level 1 and individuals at level 2. The restricted cubic splines split the trajectories into piecewise functions of age separated by "knots." Between the adjacent knots, separate cubic polynomials were fitted, with the spline terms constrained to be linear in the 2 tails. The number of knots (using the default knot positions as proposed by Harrell [31]) was chosen based on the Bayesian information criterion (BIC), with a lower BIC indicating a better fitting model. Once the best fitting model was identified, sex was added as a fixed effect and as interaction terms with the age terms identified in the previous step. Finally, an adjustment for level 1 variation was included to allow for differing error associated with measured versus self-reported BMI. From these models, fitted annual BMI values between 10 and 40 years were obtained for each individual.

Using these fitted BMI values, z-scores were created relative to the International Obesity Task Force (IOTF) reference [32]. Obesity was defined as a z-score of >2.288 in males and >2.192 in females, which corresponds to a BMI value of 30 kg/m² at 18 years. Using the sexspecific obesity cut-off, several obesity parameters were derived for each individual. Firstly, the presence of obesity at any timepoint was identified, representing any BMI z-score, which exceeded the obesity threshold. Secondly, the "number of times obese" was calculated as the number of times an individual's BMI z-score crossed upwards through the obesity threshold. Thirdly, "age first obese" was derived, representing the age, in years, when BMI z-score first crossed upwards through the obesity threshold. "Total duration of obesity" was calculated as the length of time, in years, that a person's BMI z-score exceeded the obesity threshold; these values were categorised as 0: never obese; 1: obesity 0.01 to <5 years; 2: obesity 5 to <10 years; 3: obesity 10 to <15 years; 4: obesity 15 to <20 years; 5: obesity 20+ years. Finally, we used the composite trapezoid rule to derive a cumulative obesity severity variable, represented in Fig 1 by the AUC and above the obesity threshold. Severity here is expressed in BMI-years above the obesity threshold, reflecting the fact that it incorporates both duration of obesity and the extent



BMI< obesity cut-off</p>

BMI> obesity cut-off

Fig 1. Example obesity traits (onset, duration, and severity (AUC and above obesity cut-off)) derived from the BMI-z-score trajectories of 2 random participants. AUC, area under the curve; BMI, body mass index.

https://doi.org/10.1371/journal.pmed.1003387.g001

to which BMI exceeded the age-specific obesity threshold. If this is then divided by obesity duration, it can be interpreted as the "average obesity severity," i.e., the mean excess BMI above the obesity cut-off.

Linking obesity parameters to cardiometabolic disease risk factors. Prespecified constants were added to the cardiometabolic disease risk factors to adjust for being on medication, which has been found to reduce bias [33,34]. The constants were +10 mmHg and +5 mmHg for SBP and DBP, respectively, -5% for HDL-C and +1% (absolute) for HbA1c, obtained from meta-analyses of the effect of blood pressure lowering [35], lipid-regulating [36-38], and diabetes [39] medications on the respective cardiometabolic risk factors.

Multiple linear regression was used to relate obesity parameters to the continuous cardiometabolic risk factors. As uncertainty in estimated obesity parameters are not taken into account in the confidence intervals for their associations with these continuous cardiometabolic risk factors, standard errors may be underestimated. To correct for this, robust standard errors in these subsequent models were estimated. Data were pooled across cohorts and sexes, thus enabling adjustment of the association between obesity duration and cardiometabolic risk factors for cohort and sex. As HDL-C and HbA1c required transformation to achieve normal distributions, for consistency, we transformed all continuous cardiometabolic risk factors to the 100 loge scale, so that the regression coefficients are in units of percentage difference in cardiometabolic risk factor per unit difference in covariate [40]. In a first set of models, the binary variable ever (versus never) obese (between 10 and 40 years) was tested for association with each cardiometabolic risk factor. In a subsequent set of models, we related the categorical obesity duration variable to each cardiometabolic risk factor, with never obese the referent group. The above steps were unadjusted for covariates. A subsequent model included adjustments for sex, cohort, birth weight (kg), ethnicity (white versus non-white), social class in childhood (father's social class reported when the child was 10 to 11 years and according to the Registrar General's Social Classes schema; see S2 Text for more details), and age at follow-up. A final model also included an adjustment for average obesity severity. In order to address missingness in covariate data, we used multiple imputation by chained equations (MICE) [41] combining estimates using Rubin's rules [42]. The number of imputations required to achieve convergence of parameter estimates was determined as 100 × fraction missing information (FMI) [43].

In addition, in order to aid presentation, we repeated the above steps for a number of derived dichotomous cardiometabolic disease risk factor variables, using generalised linear models (Poisson distribution with robust error variances) to estimate relative risks (RRs) for each outcome. The derived cardiometabolic disease risk factor variables were hypertension (SBP > 140mmHg and/or DBP > 90mmHg or reported use of BP-lowering medication), low HDL-cholesterol (<1.03mmol/L in males and <1.29mmol/L in females [44] or reported use of lipid-regulating medication) and elevated HbA1c (>5.7% [45] or reported use of diabetes medication).

Beta coefficients from these regression models, i.e., percentage change for continuous variables and RRs for binary variables, were plotted. Each figure is split into 2, with the left-hand side (model 1) showing the estimates from the regression of ever obese (versus never) and the right-hand side (model 2) showing estimates of the categorical obesity duration variable (versus never).

Sensitivity analyses. First, we repeated the analyses excluding the NSHD cohort as the biomedical sweep occurred much later in this cohort compared with NCDS and BCS70 cohorts, which may have resulted in an underestimation of the association between obesity duration and cardiometabolic disease risk factors. In a related sensitivity analyses, we also replaced the NSHD blood pressure variables to those collected at the 43-year sweep in order to

align with the timing of blood pressure measurements in the NCDS and BCS70. No other outcome data were available at that age in NSHD however. To identify the extent to which relationships were sex-specific, we also repeated the analyses including a "sex X obesity duration" interaction. We also performed an analysis, which was restricted to those who remained obese, assuming that relationships would strengthen when limited to those with persistent obesity and not cycles of obesity.

Analyses were performed in Stata version 15 (Stata Corp, College Station, Texas) and R version 3.5.3 (R Core Team 2019, Vienna, Austria).

This study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (<u>S1 Checklist</u>).

Code availability

The statistical code for the analyses in this paper has been placed in GitHub, the open-access online repository (repository URL: https://github.com/tomnorris1988/Obesity-duration-and-cardiometabolic-outcomes).

Results

Descriptive statistics of the cohorts are shown in Table 1. A total of 49.1% of the sample were male, and 97.2% were white British. As shown in Table 1, the prevalence of "ever obese" between 10 and 40 years was approximately 3 times higher in the most recent cohort BCS70 (19.7%, n = 1673), compared with the oldest cohort NSHD (6.6%, n = 197). The average age of first obesity onset was less in more recent cohorts, with a median of 30.2 years (interquartile range [IQR]: 25.2, 34.1) in the BCS70 compared with 33.4 years (IQR: 27.6, 37.0) in the NSHD. Accordingly, the duration of obesity was greater in the most recent cohort BCS70: median 9.7 years (IQR: 5.9, 14.7), compared with NSHD: 6.2 years (IQR: 2.7, 11.8). The negative correlation between age of obesity onset and duration of obesity was almost perfectly colinear in the more recent cohort (BCS: -0.99; NCDS: -0.95; NSHD: -0.81), indicating almost universal persistence of obesity following its onset in BCS70. Fig 1 provides examples of the BMI-z-score trajectories and the derived obesity parameters.

The average BMI at the biomedical sweep was in the overweight category (>25 kg/m²) in all 3 cohorts but was highest in the BCS70 cohort (27.6 kg/m²; IQR: 24.6, 31.5). For all 5 cardiometabolic risk factors, the mean was highest in the NSHD cohort, reflecting the older age at follow-up. This was most notable for SBP, with a mean of 136.0 mmHg (SD: 20.1) in the NSHD compared with 126.6 mmHg (16.5) and 124.6 mmHg (15.2) in the NCDS and BCS70, respectively. This translated to a much higher prevalence of hypertension in the NSHD cohort (68.1%) compared with the NCDS (27.8%) and BCS70 cohorts (23.9%). The presence of elevated HbA1c was also considerably higher in the NSHD cohort compared with the NCDS and BCS70 (35.8% versus 15.0% and 16.5%, respectively).

Relationship of obesity parameters with cardiometabolic disease risk factors

Results from the unadjusted analysis are included in <u>S1</u> and <u>S2</u> Tables. Here, we report estimates from the adjusted analyses, presented in Figs 2-4 and with corresponding estimates in <u>S3-S6</u> Tables.

HbA1c. Being ever obese at any age between 10 and 40 years (versus never obese) was associated with a 9.0% (95% CI: 8.2, 9.9) higher HbA1c (Fig 2, left panel), which reduced to 4.5% higher (95% CI: 3.5, 5.6) when adjusted for obesity severity. HbA1c increased linearly with obesity duration, from 5% excess for <5 years duration up to 19.9% (95% CI: 16.5, 23.3) for 20 to 30 years duration (p(trend) < 0.001). Upon adjustment for obesity severity, the trend

		NSHD 1946 (<i>n</i> = 2,968)		1958 NCDS ($n = 9,302$)		1970 BCS (<i>n</i> = 8,476)	
Sex							
Males	n (%)	1,459 (49.2)		4,630 (49.8)		4,106 (48.4)	
Females	n (%)	1,509 (50.8)		4,672 (50.2)		4,370 (51.6)	
Ethnicity							
White British	n (%)	2,968 (100)		9,089 (97.7)		7,882 (93.0)	
Other ^a	n (%)	0 (0)		205 (2.2)		376 (4.4)	
Missing	n (%)	0 (0)		8 (0.1)		218 (2.6)	
Obesity traits							
Never obese	n (%)	2,771 (93.4)		8,267 (88.9)		6,803 (80.3)	
Ever obese	n (%)	197 (6.6)		1,035 (11.1)		1,673 (19.7)	
Age first onset (years)	Median (IQR)	33.4 (27.6; 37.0)		31.5 (25.4; 36.1)		30.2 (25.2; 34.1)	
Total duration (years)	Median (IQR)	6.2 (2.7; 11.8)		8.3 (3.9; 14.4)		9.7 (5.9; 14.7)	
Correlation (age onset × duration obese)		-0.81		-0.95		-0.99	
Number of periods							
1	n (%)	192 (97.5)		1,023 (98.8)		1,671 (99.9)	
2	n (%)	4 (2.0)		12 (1.2)		2 (0.1)	
3	n (%)	1 (0.5)		0		0	
Obesity severity (BMI-years)	Median (IQR)	5.6 (1.1; 22.7)		9.4 (1.7; 35.0)		17.1 (4.3; 48.0)	
Correlation (duration × severity)		0.86		0.85		0.80	
Biomedical outcomes		% missing		% missing		% missing	
Age at follow-up (years)	Mean (SD)	-	53.5 (0.2)	-	45.2 (0.4)	-	47.3 (0.7)
BMI at follow-up (kg/m ²)	Median (IQR)	1.3	26.6 (24.2; 29.9)	1.3	26.6 (24.0; 29.9)	13.4	27.6 (24.6; 31.5)
Obese at follow-up (BMI > 30 kg/m ²)	n (%)	1.3	707 (24.1)	1.3	2,239 (24.4)	13.4	2,424 (33.0)
Systolic blood pressure (mmHg)*	Mean (SD)	1.9	136.0 (20.1)	0.5	126.5 (16.5)	11.5	124.6 (15.2)
Diastolic blood pressure (mmHg)*	Mean (SD)	1.9	84.4 (12.2)	0.5	78.8 (10.8)	11.5	77.3 (11.0)
Hypertension ^b	n (%)	1.9	1,993 (68.1)	0.5	2,578 (27.8)	11.3	1,798 (23.9)
HDL-C (mmol/L)*	Median (IQR)	20.2	1.6 (1.3; 2.0)	16.1	1.5 (1.3; 1.8)	29.5	1.5 (1.2; 1.8)
Low-HDL-C ^c	n (%)	19.3	312 (13.0)	14.5	1,595 (20.1)	28.5	1,385 (22.8)
HbA1c (%)*	Median (IQR)	13.6	5.7 (5.3; 5.9)	15.2	5.3 (5.0; 5.4)	29.9	5.4 (5.3; 5.6)
Elevated HbA1c ^d	n (%)	13.2	921 (35.8)	13.8	1,206 (15.0)	39.3	987 (16.5)

Table 1. Descriptive statistics for life course obesity parameters and cardiometabolic disease risk factors at the biomedical sweep of those in target study sample (n = 20,746).

* Original values (i.e., not adjusted for medication use).

a Other ethnicities: white other, mixed race, Indian, Pakistani, Bangladeshi, Other Asian, Caribbean, African, other black, and Chinese.

b Hypertension: SBP/DBP \geq 140/90 mmHg and/or on BP-lowering medication.

c Low-HDL-C: according to NCEP ATPIII criteria and/or on lipid-regulating medication.

d Elevated HbA1c: according to CDC criteria and/or on diabetes medication.

BCS, 1970 British Cohort Study; BMI, body mass index; CDC, Centers for Disease Control and Prevention; DBP, diastolic blood pressure; HbA1c, glycated haemoglobin; HDL-C, high-density-lipoprotein cholesterol; NCDS, National Child Development Study; NCEP ATPIII, Adult Treatment Panel III (ATP III) report of the National Cholesterol Education Program (NCEP); NSHD, National Survey of Health and Development; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1003387.t001

remained (p(trend) = 0.007) but was attenuated, particularly for 20 to 30 years, which reduced from 19.9% to 11.6% (95% CI: 5.9, 17.2), a relative reduction of 42%.

There was also a linear trend between obesity duration and risk for elevated HbA1c, with those obese for <5 years having a 2.1 (95% CI: 1.8, 2.4) times higher risk of elevated HbA1c compared with never obese, which more than doubled in those obese for 20 to 30 years (RR 4.6; 95% CI: 3.9, 5.5, p(trend) < 0.001) (Fig 2, right panel). However, upon adjustment for obesity severity, this graded relationship was attenuated (p(trend) = 0.006).



Fig 2. Association between ever obese and categories of obesity duration (versus never obese) and HbA1c (left panel) and risk for elevated HbA1c (right panel).

https://doi.org/10.1371/journal.pmed.1003387.g002

SBP and DBP. There was a positive relationship between ever being obese between 10 and 40 years and both SBP and DBP. For example, ever obese was associated with a 6.1% (95% CI: 5.6, 6.6) higher SBP and 7.1% (95% CI: 6.6, 7.7) higher in DBP at follow-up (versus never obese) (Fig 3A and 3B). Obesity duration was also positively associated with both SBP and DBP, such that SBP was 5.0% higher in those who were obese for <5 years compared with those never obese, increasing to 9.0% higher for 20 to 30 years (p(trend) < 0.001). However, upon adjustment for obesity severity, evidence for this dose-response association was greatly reduced (SBP: p(trend) = 0.975; DBP: p(trend) = 0.294).



Fig 3. Association between ever obese and categories of obesity duration (versus never obese) and SBP, DBP, and HDL-C. DBP, diastolic blood pressure; HDL-C, high-density-lipoprotein cholesterol; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1003387.g003





https://doi.org/10.1371/journal.pmed.1003387.g004

Consistent with these findings, ever being obese between 10 and 40 years (versus never) was associated with an RR for hypertension of 1.6 (95% CI: 1.5, 1.7), independent of obesity severity (Fig 4A and S6 Table). For obesity duration, a similar pattern was observed to that seen for SBP and DBP, i.e., a gradually increasing risk for hypertension with increasing time spent obese (p(trend) < 0.001), evidence for which weakened when adjusted for obesity severity (p(trend) = 0.456).

HDL-C. A negative relationship was observed between obesity and HDL-C, such that obesity at any point between 10 and 40 years was associated with a 16.4% (95% CI: 17.6, 15.2) lower HDL-C at follow-up (Fig 3C), attenuating to 12.3% lower when adjusted for severity. There was a linear trend in the effect of obesity duration on HDL-C, such that HDL-C levels in those with obesity for <5 years were 12.4% (95% CI: 10.4, 14.4) lower than those never obese, which increased to 24.8% (95% CI: 20.5, 29.1) lower in those who had been obese for 20 to 30 years (*p*(trend) < 0.001). Upon adjustment for obesity severity, evidence for the trend attenuated (*p*(trend) = 0.117).

This resulted in an RR for low-HDL-C of 2.0 (95% CI: 1.8, 2.2) in those who were ever obese between 10 and 40 years (versus never), independent of obesity severity (Fig 4B and S6 Table). For obesity duration, there was a linear trend of increasing risk (p(trend) < 0.001), which remained on adjustment for severity, although evidence for this was attenuated (p(trend) = 0.037).

Sensitivity analysis. Similar results were found when the analysis was limited to the NCDS and BCS70 cohorts (<u>S7</u> and <u>S8</u> Tables), thus accounting for the difference in the age at follow-up in the NSHD. Similarly, replacing the blood pressure variables in the NSHD cohort with those collected at the age 43-year sweep, in order to be more aligned with the age at follow-up in the NCDS and BCS70, did not change results (<u>S9</u> and <u>S10</u> Tables). When stratified by sex, associations were consistently stronger in females (<u>S11</u> and <u>S12</u> Tables) and especially for the dichotomous cardiometabolic disease risk factor variables. Finally, estimates were largely unchanged when the analysis was limited to those with persistent obesity (i.e., staying obese after first onset) (<u>S13</u> and <u>S14</u> Tables).

Discussion

This study utilised longitudinal BMI data from 3 British birth cohort studies to model each person's obesity history and derive individual obesity parameters. Ever being obese between 10 and 40 years of age, compared with never being obese, was associated with less favourable levels of all cardiometabolic disease risk factors. More time spent obese was associated with worse profiles for all cardiometabolic disease risk factors, although greatest for HbA1c. When adjustment was made for obesity severity, the strength of the evidence in support of an association between obesity duration and SBP, DBP, and HDL-C was weak (p > 0.1). For HbA1c however, although the association with obesity duration also attenuated when adjusting for obesity severity, the strength of evidence remained strong. The study design, in particular the fact that most individuals who became obese remained obese, has meant that age of obesity onset and obesity duration are very highly negatively correlated. Our results also therefore mean that after accounting for obesity severity, an earlier age of obesity onset was only associated with HbA1C. These key findings were robust to a range of sensitivity analyses.

In attempting to separate the effects of obesity duration and severity on cardiometabolic health, previous studies have simply adjusted for BMI (or waist circumference) at the time of outcome assessment [14,15,19,21,22,46]. This, however, only provides an indication of obesity severity at that particular point in time. Our study represents an advance over these studies however, as we have been able to measure obesity severity accumulated over the life course, and by adjusting this for the time spent obese, we have been able to appropriately separate the effects of obesity duration and severity. As such, these findings provide novel, robust evidence regarding the independent association of obesity duration with cardiometabolic disease risk factors.

Our findings are in line with other studies, which have observed an attenuated, but persisting, effect of obesity duration on diabetes risk or impaired glucose metabolism, once obesity severity is accounted for [15,19,21,22]. In another NCDS analysis (n = 7855), Power and colleagues [19] observed that compared with those never obese, those with the greatest duration of obesity (i.e., onset < 16 years) had an almost 24-fold increased risk of having HbA1c of >7% (and/or a diagnosis of diabetes) at 45 years. While this risk was substantially attenuated upon adjustment for current BMI, it still remained over 4 times higher compared with those never obese. In addition we have observed, in line with Pontirolli and colleagues [15], a specific effect of obesity duration on glucose metabolism. In their study of 760 obese adults (average age 51 years), obesity duration was a risk factor for glucose intolerance and type 2 diabetes but not for hypertension or hyperlipidaemia [15]. Evidence in support of our finding of no independent association of obesity duration with HDL-C is lacking. To our knowledge only 1 other study has investigated this and observed an association in females only, although the strength of evidence was modest (p = 0.05) [14].

In addition to the cited empirical studies, there is also a plausible biological mechanism supporting the observed association between obesity duration and HbA1c (reflecting impaired glucose metabolism). Obesity is characterised by enlarged fat stores, which results in enhanced lipolysis and an increase in circulating free fatty acids. This state leads to peripheral and hepatic insulin resistance [47,48], resulting in a compensatory insulin hypersecretion by the pancreatic β -cells in order to preserve normoglycemia [49]. Prolonged obesity leads to β -cell exhaustion [50], culminating in a reduced insulin response and an inability to maintain normoglycemia [51]. In addition, prolonged obesity may represent a state in which subcutaneous adipose stores have been exhausted, with the consequence being a deposition of adipose tissue around the visceral organs (e.g., liver and pancreas), with fat stored in these areas (i.e., "ectopic fat") being strongly related to insulin resistance [52]. Despite the persisting independent effect of obesity duration on HbA1c levels, a substantial reduction in the effect was observed once the severity of obesity had been accounted for. This suggests that in those who have been exposed to obesity for a prolonged period, there is still opportunity to return to more favourable HbA1c levels if a degree of weight loss is achieved. For example, upon adjustment for severity, the risk of elevated HbA1c in those who had been obese for 20 to 30 years reduced from more than a 4-fold increased risk (relative to never obese), to a level similar to those obese for half as long, i.e., 10 to 15 years (RR: 3.0; 95%CI: 2.3, 4.0).

There was some evidence that the association between obesity duration and the dichotomous cardiometabolic outcomes was stronger in females than in males (S11 and S12 Tables). Sex-specific associations have been observed in other studies [14,22,53]. Janssen and colleagues [14], for example, observed an independent effect of overweight/obesity duration on risk for insulin resistance and type 2 diabetes (and also hypertension, hypertriglyceridemia, low-HDL-C, and metabolic syndrome) in females, but not in males except for hypertriglyceridemia. A sex difference was also observed in the Framingham Heart Study [53]. Tanamas and colleagues observed an association between obesity duration and risk for hypertension in females but not in males (ages 30 to 62 years) [53]. As summarised in the review by Jarvis [54], there are fundamental differences in the control of metabolic homeostasis between males and females. Females are more likely to gain fat, and although abdominal obesity more commonly affects males than females, the prevalence of abdominal obesity has increased more in females than in males [55]. Furthermore, the prevalence of visceral obesity associated with metabolic syndrome is 2 to 10 times higher in women throughout the world [56-58]. It may be therefore that compared with males, females are more exposed to this metabolically volatile adipose tissue and thus at increased risk of its deleterious outcomes.

Strengths

The key strength of our study is the derivation, using over 130,000 serial BMI observations across the life course, of individualised obesity parameters, which enabled us to distinguish between obesity severity and duration. In addition, the pooling of data from 3 nationally representative cohorts means the observed associations are based on a far larger sample than most previous studies and are likely to be generalisable to the underlying population.

Limitations

Our definition of obesity was based on BMI, which despite exhibiting a strong positive correlation with direct estimates of fat mass [59], is only an indicator of total body adiposity. However, it remains the most commonly used, widely accepted and practical measure of obesity in both children and adults. Our trajectories were dependent on the frequency of BMI measurements across the life course, with some intervals spanning 10 years. As such, we may not have captured instances of weight cycling between measurement occasions. Measurement protocols for weight and height were not consistent within and between studies, which may have introduced bias if self-reported measurements were systemically under or overreported. It has been shown that people with greater BMIs tend to underreport their weight [60,61], suggesting that estimates of obesity duration (and severity) may be conservative in our study. Our regression models included adjustment for only a small number of covariates, which means there is a possibility of residual confounding. As we have combined 3 cohorts, any included variable must be harmonised across each cohort so that the variable conveys the same thing in each cohort. This is only the case for a small number of variables in the cohorts we have used. As all of the included studies suffered from attrition, which is more extensive in those from lower socioeconomic position (SEP) groups and/or with higher BMI [62,63], we may have inadvertently selected a more socioeconomically advantaged and thinner sample which, in addition to a loss of power, may have introduced bias into the observed associations. In addition, as the NSHD, NCDS, and BCS70 cohorts are either exclusively (NSHD), or predominantly white British, we are unable to generalise the results to other ethnic groups. Finally, the biomedical sweep in the NSHD cohort was conducted 9 and 7 years later than the NCDS and BCS cohorts, respectively, which may impair cross-cohort comparability (underpinning the decision to pool cohorts). However, supplementary analyses limited to the NCDS and BCS cohorts only and replacing the NSHD blood pressure variables with those collected at 43 years produced similar estimates (S7–S10 Tables).

Associations observed in this study suggest that there are benefits in delaying the onset of obesity, as risks of elevated HbA1c were positively associated with time spent with obesity, independent of the degree of severity. Interventions aiming to prevent childhood obesity therefore have the potential to reduce the long-term risk of developing diabetes. However, we also observed an amelioration of HbA1c profiles in those who had been exposed to obesity for a prolonged period, once severity of obesity is accounted for. As such, people with obesity should be encouraged to lose weight in order to return their HbA1c levels to more favourable values. Firstly, however, more research using different epidemiological approaches underpinned by different assumptions and sources of bias (e.g., mendelian randomization and negative control studies) are needed to test the robustness of these findings.

Conclusion

We found a dose–response relationship between the duration of obesity and HbA1c, independent of obesity severity. Given that the obesity epidemic is characterised by trends towards earlier onset and consequently greater lifetime exposure, our findings are important as they suggest that health policy recommendations aimed at preventing early onset obesity, and therefore reducing lifetime obesity exposure, may help reduce the risk for diabetes. For those who are already obese, reducing obesity severity can also improve their metabolic profile. Accordingly, prevention strategies could consider both the duration and severity of obesity.

Supporting information

S1 Checklist. STROBE checklist. (DOCX)

S1 Text. Measurement protocol for collection of cardiometabolic outcomes in adulthood. (DOCX)

S2 Text. Childhood social class. (DOCX)

S1 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors*† (imputed, unadjusted). (DOCX)

S2 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, unadjusted). (DOCX)

S3 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors*† (imputed, adjusted for sex, cohort, age at

follow-up, ethnicity, birth weight, and childhood social class). (DOCX)

S4 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, and childhood social class). (DOCX)

S5 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors*† (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity). (DOCX)

S6 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity). (DOCX)

S7 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): excluding NSHD.

```
(DOCX)
```

S8 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): excluding NSHD.

(DOCX)

S9 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): using blood pressure at 43 years in NSHD.

(DOCX)

S10 Table. Association between ever obese and categories of obesity duration (versus never obese) and categorical cardiometabolic outcomes (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): using blood pressure at 43 years in NSHD. (DOCX)

S11 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors (imputed, adjusted for cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): sex interaction.

(DOCX)

S12 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, adjusted for cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): sex interaction.

(DOCX)

S13 Table. Association between ever obese and categories of obesity duration (versus never obese) and cardiometabolic disease risk factors (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): limited to those who once obese were always obese. (DOCX)

S14 Table. Association between ever obese and categories of obesity duration (versus never obese) and dichotomous cardiometabolic outcomes (imputed, adjusted for sex, cohort, age at follow-up, ethnicity, birth weight, childhood social class, and obesity severity): limited to those who once obese were always obese.

(DOCX)

S1 Fig. Sample flow diagram. (TIFF)

Author Contributions

Conceptualization: Tom Norris, Tim J. Cole, Russell Viner, William Johnson.

Data curation: Mark Hamer, Rebecca Hardy, George B. Ploubidis, William Johnson.

Formal analysis: Tom Norris, Tim J. Cole.

Funding acquisition: William Johnson.

Methodology: Tom Norris, David Bann, Mark Hamer, Rebecca Hardy, Leah Li, Ken K. Ong, George B. Ploubidis, Russell Viner, William Johnson.

Supervision: William Johnson.

Writing - original draft: Tom Norris.

Writing – review & editing: Tom Norris, Tim J. Cole, David Bann, Mark Hamer, Rebecca Hardy, Leah Li, Ken K. Ong, George B. Ploubidis, William Johnson.

References

- Non-Communicable Diseases Risk Factor Collaboration. Worldwide trends in body-mass index, underweight, overweight, and obesity from 1975 to 2016: a pooled analysis of 2416 population-based measurement studies in 128-9 million children, adolescents, and adults. Lancet. 2017; 390:2627–42. https://doi.org/10.1016/S0140-6736(17)32129-3 PMID: 29029897
- Global Burden of Disease 2015 Obesity Collaborators. Health effects of overweight and obesity in 195 countries over 25 years. N Engl J Med. 2017; 377:13–27. https://doi.org/10.1056/NEJMoa1614362 PMID: 28604169
- Johnson W. Body size trajectories and cardio-metabolic resilience to obesity. Nutr Bull. 2018; 43:456– 62. https://doi.org/10.1111/nbu.12350 PMID: 30555276
- 4. Johnson W. Healthy obesity: time to give up the ghost? Annals of Human Biology. Taylor and Francis Ltd. 2018:297–8. https://doi.org/10.1080/03014460.2018.1444789 PMID: 29641913
- Johnson W, Li L, Kuh D, Hardy R. How Has the Age-Related Process of Overweight or Obesity Development Changed over Time? Co-ordinated Analyses of Individual Participant Data from Five United Kingdom Birth Cohorts. PLoS Med. 2015; 12. https://doi.org/10.1371/journal.pmed.1001828 PMID: 25993005
- Abraham S, Collins G, Nordsieck M. Relationship of childhood weight status to morbidity in adults. HSMHA Health Rep. 1971; 86: 273–84. Available: http://www.ncbi.nlm.nih.gov/pubmed/5547793 PMID: 5547793
- Suglia SF, Clark CJ, Gary-Webb TL. Adolescent obesity, change in weight status, and hypertension: Racial/ethnic variations. Hypertension. 2013; 61:290–5. https://doi.org/10.1161/HYPERTENSIONAHA. 111.00214 PMID: 23248147

- Bjerregaard LG, Jensen BW, Ängquist L, Osler M, Sørensen TIA, Baker JL. Change in overweight from childhood to early adulthood and risk of type 2 diabetes. N Engl J Med. 2018; 378:1302–12. <u>https://doi. org/10.1056/NEJMoa1713231</u> PMID: 29617589
- Juonala M, Magnussen CG, Berenson GS, Venn A, Burns TL, Sabin MA, et al. Childhood Adiposity, Adult Adiposity, and Cardiovascular Risk Factors. N Engl J Med. 2011; 365:1876–85. <u>https://doi.org/10.1056/NEJMoa1010112 PMID: 22087679</u>
- Juonala M, Raitakari M, SA J, Raitakari OT. Obesity in youth is not an independent predictor of carotid IMT in adulthood: The Cardiovascular Risk in Young Finns Study. Atherosclerosis. 2006; 185:388–93. https://doi.org/10.1016/j.atherosclerosis.2005.06.016 PMID: 16045913
- Mongraw-Chaffin M, Foster MC, Kalyani RR, Vaidya D, Burke GL, Woodward M, et al. Obesity severity and duration are associated with incident metabolic syndrome: Evidence against metabolically healthy obesity from the multi-ethnic study of atherosclerosis. J Clin Endocrinol Metab. 2016; 101:4117–24. https://doi.org/10.1210/jc.2016-2460 PMID: 27552544
- The NS, Richardson AS, Gordon-Larsen P. Timing and duration of obesity in relation to diabetes: Findings from an ethnically diverse, nationally representative sample. Diabetes Care. 2013; 36:865–72. https://doi.org/10.2337/dc12-0536 PMID: 23223352
- Moran A, Jacobs DR, Steinberger J, Hong CP, Prineas R, Luepker R, et al. Insulin resistance during puberty: Results from clamp studies in 357 children. Diabetes. 1999; 48:2039–44. <u>https://doi.org/10. 2337/diabetes.48.10.2039 PMID: 10512371</u>
- Janssen I, Katzmarzyk PT, Ross R. Duration of overweight and metabolic health risk in American men and women. Ann Epidemiol. 2004; 14:585–91. https://doi.org/10.1016/j.annepidem.2004.01.007 PMID: 15350959
- Pontiroli AE, Galli L. Duration of obesity is a risk factor for non-insulin-dependent diabetes mellitus, not for arterial hypertension or for hyperlipidaemia. Acta Diabetol. 1998; 35:130–6. <u>https://doi.org/10.1007/</u> s005920050117 PMID: 9840448
- Wills AK, Hardy RJ, Black S, Kuh DJ. Trajectories of overweight and body mass index in adulthood and blood pressure at age 53: The 1946 British birth cohort study. J Hypertens. 2010; 28:679–86. <u>https:// doi.org/10.1097/HJH.0b013e328335de7b PMID: 20042875</u>
- Wannamethee SG, Shaper AG. Weight change and duration of overweight and obesity in the incidence of type 2 diabetes. Diabetes Care. 1999; 22:1266–72. <u>https://doi.org/10.2337/diacare.22.8.1266</u> PMID: 10480769
- Wannamethee SG, Shaper AG, Walker M. Overweight and obesity and weight change in middle aged men: Impact on cardiovascular disease and diabetes. J Epidemiol Community Health. 2005; 59:134–9. https://doi.org/10.1136/jech.2003.015651 PMID: 15650145
- Power C, Thomas C. Changes in BMI, duration of overweight and obesity, and glucose metabolism: 45 Years of follow-up of a birth cohort. Diabetes Care. 2011; 34:1986–91. <u>https://doi.org/10.2337/dc10-1482</u> PMID: 21775760
- Reis JP, Hankinson AL, Loria CM, Lewis CE, Powell-Wiley T, Wei GS, et al. Duration of abdominal obesity beginning in young adulthood and incident diabetes throughmiddle age: The CARDIA study. Diabetes Care. 2013; 36:1241–7. https://doi.org/10.2337/dc12-1714 PMID: 23248193
- Hu Y, Bhupathiraju SN, De Koning L, Hu FB. Duration of obesity and overweight and risk of type 2 diabetes among US women. Obesity. 2014; 22:2267–73. <u>https://doi.org/10.1002/oby.20851</u> PMID: 25131512
- 22. Abdullah A, Stoelwinder J, Shortreed S, Wolfe R, Stevenson C, Walls H, et al. The duration of obesity and the risk of type 2 diabetes. Public Health Nutr. 2011; 14:119–26. <u>https://doi.org/10.1017/S1368980010001813 PMID: 20587115</u>
- Everhart JE, Pettitt DJ, Bennett PH, Knowler WC. Duration of obesity increases the incidence of NIDDM. Diabetes. 1992; 41:235–40. https://doi.org/10.2337/diab.41.2.235 PMID: 1733815
- Abdullah A, Wolfe R, Mannan H, Stoelwinder JU, Stevenson C, Peeters A. Epidemiologic merit of obese-years, the combination of degree and duration of obesity. Am J Epidemiol. 2012; 176:99–107. https://doi.org/10.1093/aje/kwr522 PMID: 22759723
- Araújo J, Severo M, Barros H, Ramos E. Duration and degree of adiposity: Effect on cardiovascular risk factors at early adulthood. Int J Obes. 2017; 41:1526–30. https://doi.org/10.1038/ijo.2017.133 PMID: 28584300
- Wadsworth M, Kuh D, Richards M, Hardy R. Cohort Profile: The 1946 National Birth cCohort (MRC National Survey of Health and Development). Int J Epidemiol. 2006; 35:49–54. <u>https://doi.org/10.1093/</u> ije/dyi201 PMID: 16204333
- Power C, Elliott J. Cohort profile: 1958 British birth cohort (National Child Development Study). Int J Epidemiol. 2006; 35:34–41. https://doi.org/10.1093/ije/dyi183 PMID: 16155052

- Elliott J, Shepherd P. Cohort Profile: 1970 British Birth Cohort (BCS70). Int J Epidemiol. 2006; 35:836– 43. https://doi.org/10.1093/ije/dyl174 PMID: 16931528
- Kuh D, Pierce M, Adams J, Deanfield J, Ekelund U, Friberg P, et al. Cohort profile: updating the cohort profile for the MRC National Survey of Health and Development: a new clinic-based data collection for ageing research. Int J Epidemiol. 2011; 40:e1–9. https://doi.org/10.1093/ije/dyq231 PMID: 21345808
- Johnson W, Li L, Kuh D, Hardy R. How Has the Age-Related Process of Overweight or Obesity Development Changed over Time? Co-ordinated Analyses of Individual Participant Data from Five United Kingdom Birth Cohorts PLOS Med. 2015; 12:e1001828. https://doi.org/10.1371/journal.pmed.1001828 PMID: 25993005
- Harrell F. Regression Modelling Strategies: With applications to Linear Models, Logistic Regression, and Survival Analysis. Springer-Verlag; 2001.
- Cole TJ, Bellizzi MC, Flegal KM, Dietz WH. Establishing a standard definition for child overweight and obesity worldwide: International survey. Br Med J. 2000; 320:1240–3. <u>https://doi.org/10.1136/bmj.320.</u> 7244.1240 PMID: 10797032
- White IR, Koupilova I, Carpenter J. The use of regression models for medians when observed outcomes may be modified by interventions. Stat Med. 2003:1083–96. <u>https://doi.org/10.1002/sim.1408</u> PMID: 12652555
- Tobin MD, Sheehan NA, Scurrah KJ, Burton PR. Adjusting for treatment effects in studies of quantitative traits: Antihypertensive therapy and systolic blood pressure. Stat Med. 2005; 24:2911–35. https:// doi.org/10.1002/sim.2165 PMID: 16152135
- Law MR, Wald NJ, Morris JK, Jordan RE. Value of low dose combination treatment with blood pressure lowering drugs: Analysis of 354 randomised trials. Br Med J. 2003; 326:1427–31. https://doi.org/10. 1136/bmj.326.7404.1427 PMID: 12829555
- Wierzbicki AS, Mikhailidis DP. Dose-response effects of atorvastatin and simvastatin on high-density lipoprotein cholesterol in hypercholesterolaemic patients: A review of five comparative studies. Int J Cardiol. 2002; 84:53–7. https://doi.org/10.1016/s0167-5273(02)00118-3 PMID: 12104065
- Costa J, Borges M, David C, Vaz CA. Efficacy of lipid lowering drug treatment for diabetic and non-diabetic patients: Meta-analysis of randomised controlled trials. Br Med J. 2006; 332:1115–8. <u>https://doi.org/10.1136/bmj.38793.468449.AE PMID: 16585050</u>
- Ki M, Pouliou T, Li L, Power C. Physical (in)activity over 20y in adulthood: Associations with adult lipid levels in the 1958 British birth cohort. Atherosclerosis. 2011; 219:361–7. <u>https://doi.org/10.1016/j.</u> atherosclerosis.2011.07.109 PMID: 21855876
- Bennett WL, Maruthur NM, Singh S, Segal JB, Wilson LM, Chatterjee R, et al. Comparative effectiveness and safety of medications for type 2 diabetes: An update including new drugs and 2-drug combinations. Annals of Internal Medicine American College of Physicians. 2011:602–18. <u>https://doi.org/10.</u> 7326/0003-4819-154-9-201105030-00336 PMID: 21403054
- Cole TJ. Sympercents: Symmetric percentage differences on the 100 log(e) scale simplify the presentation of log transformed data. Stat Med. 2000; 19:3109–25. https://doi.org/10.1002/1097-0258 (20001130)19:22<3109::aid-sim558>3.0.co;2-f PMID: 11113946
- 41. Royston P, White I. Multiple Imputation by Chained Equations (MICE): Implementation in Stata. J Stat Softw. 2011;45. https://doi.org/10.18637/jss.v045.i04
- 42. Rubin D. Multiple imputation for nonresponse in surveys. John Wiley & Sons, Ltd; 1987. Available: https://books.google.co.uk/books?hl=en&Ir=&id=bQBtw6rx_mUC&oi=fnd&pg=PR24&dq=multiple +imputation+for+nonresponse+in+surveys&ots=80pM8KZ0hQ&sig=UxYMFHtf1xNVr_ m85V6Bmv5iETY
- **43.** White IR, Royston P, Wood AM. Multiple imputation using chained equations: Issues and guidance for practice. Stat Med. 2011; 30:377–99. https://doi.org/10.1002/sim.4067 PMID: 21225900
- 44. National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation and T of HBC in A (Adult TPI). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III) final report. Circulation. 2002; 106: 3143–3421. https://doi.org/10.1097/00019616-199311000-00022 PMID: 12485966
- American Diabetes Association. Diagnosis and classification of diabetes mellitus. Diabetes Care. 2014; 37. https://doi.org/10.2337/dc14-S081 PMID: 24357215
- Reis JP, Loria CM, Lewis CE, Powell-Wiley TM, Wei GS, Carr JJ, et al. Association between duration of overall and abdominal obesity beginning in young adulthood and coronary artery calcification in middle age. JAMA—J Am Med Assoc. 2013; 310:280–8. <u>https://doi.org/10.1001/jama.2013.7833</u> PMID: 23860986

- Kissebah AH. Intra-abdominal fat: is it a major factor in developing diabetes and coronary artery disease? Diabetes Res Clin Pract. 1996; 30 (Suppl):25–30. https://doi.org/10.1016/s0168-8227(96) 80035-0
- Felber JP, Golay A. Pathways from obesity to diabetes. Int J Obes. 2002; 26:S39–45. <u>https://doi.org/10.1038/sj.ijo.0802126 PMID: 12174327</u>
- 49. Prentki M, Nolan CJ. Islet β cell failure in type 2 diabetes. Journal of Clinical Investigation. 2006:1802– 12. https://doi.org/10.1172/JCl29103 PMID: 16823478
- Groop LC, Eriksson JG. The etiology and pathogenesis of non-insulin-dependent diabetes. Ann Med. 1992; 24:483–9. https://doi.org/10.3109/07853899209167000 PMID: 1485943
- 51. Leahy JL. Pathogenesis of type 2 diabetes mellitus. Archives of Medical Research. 2005:197–209. https://doi.org/10.1016/j.arcmed.2005.01.003 PMID: 15925010
- Sattar N, Gill JMR. Type 2 diabetes as a disease of ectopic fat? BMC Med. 2014; 12. <u>https://doi.org/10.1186/s12916-014-0123-4 PMID: 25159817</u>
- 53. Tanamas SK, Wong E, Backholer K, Abdullah A, Wolfe R, Barendregt J, et al. Duration of obesity and incident hypertension in adults from the Framingham Heart Study. J Hypertens. 2015; 33:542–5. https://doi.org/10.1097/HJH.00000000000441 PMID: 25479024
- 54. Mauvais-Jarvis F. Sex differences in metabolic homeostasis, diabetes, and obesity. Biol Sex Differ. 2015; 6:1–9. https://doi.org/10.1186/s13293-014-0019-1 PMID: 25653823
- 55. NatCen Social Research and UCL. Health Survey for England 2018. 2019. Available from: https:// digital.nhs.uk/data-and-information/publications/statistical/health-survey-for-england/2018
- Al-Lawati JA, Mohammed AJ, Al-Hinai HQ, Jousilahti P. Prevalence of the metabolic syndrome among Omani adults. Diabetes Care. 2003; 26:1781–5. https://doi.org/10.2337/diacare.26.6.1781 PMID: 12766109
- Gupta R, Deedwania PC, Gupta A, Rastogi S, Panwar RB, Kothari K. Prevalence of metabolic syndrome in an Indian urban population. Int J Cardiol. 2004; 97:257–61. https://doi.org/10.1016/j.ijcard. 2003.11.003 PMID: 15458693
- Gu D, Reynolds K, Wu X, Chen J, Duan X, Reynolds RF, et al. Prevalence of the metabolic syndrome and overweight among adults in China. Lancet. 2005; 365:1398–405. https://doi.org/10.1016/S0140-6736(05)66375-1 PMID: 15836888
- Flegal KM, Shepherd JA, Looker AC, Graubard BI, Borrud LG, Ogden CL, et al. Comparisons of percentage body fat, body mass index, waist circumference, and waist-stature ratio in adults. Am J Clin Nutr. 2009; 89:500–8. https://doi.org/10.3945/ajcn.2008.26847 PMID: 19116329
- 60. Crawley HF, Portides G. Self-reported versus measured height, weight and body mass index amongst 16–17 year old British teenagers. Int J Obes Relat Metab Disord. 1995; 19: 579–84. Available. <u>http://</u> www.ncbi.nlm.nih.gov/pubmed/7489030. PMID: 7489030
- Shields M, Gorber SC, Tremblay MS. Estimates of obesity based on self-report versus direct measures. Heal Reports. 2008; 19:61–76. PMID: 18642520
- Stafford M, Black S, Shah I, Hardy R, Pierce M, Richards M, et al. Using a birth cohort to study ageing: representativeness and response rates in the National Survey of Health and Development. Eur J Ageing. 2013; 10:145–57. https://doi.org/10.1007/s10433-013-0258-8 PMID: 23637643
- Mostafa T, Wiggins RD. The impact of attrition and non-response in birth cohort studies: a need to incorporate missingness strategies. Longit Life Course Stud. 2015; 6:131–46. https://doi.org/10.14301/llcs.v6i2.312